

# Central Lancashire Online Knowledge (CLoK)

| Title    | Incidence, prevalence, and mortality of autoimmune hepatitis in England 1997-2015. A population-based cohort study   |
|----------|--|
| Туре     | Article  |
| URL      | https://clok.uclan.ac.uk/id/eprint/33095/  |
| DOI      | https://doi.org/10.1111/liv.14480  |
| Date     | 2020   |
| Citation | Grønbæk, Lisbet, Otete, Harmony, Ban, Lu, Crooks, Colin, Card, Timothy,<br>Jepsen, Peter and West, Joe (2020) Incidence, prevalence, and mortality of<br>autoimmune hepatitis in England 1997-2015. A population-based cohort<br>study. Liver International, 40 (7). pp. 1634-1644. ISSN 1478-3223 |
| Creators | Grønbæk, Lisbet, Otete, Harmony, Ban, Lu, Crooks, Colin, Card, Timothy,<br>Jepsen, Peter and West, Joe   |

It is advisable to refer to the publisher's version if you intend to cite from the work. https://doi.org/10.1111/liv.14480

For information about Research at UCLan please go to <a href="http://www.uclan.ac.uk/research/">http://www.uclan.ac.uk/research/</a>

All outputs in CLoK are protected by Intellectual Property Rights law, including Copyright law. Copyright, IPR and Moral Rights for the works on this site are retained by the individual authors and/or other copyright owners. Terms and conditions for use of this material are defined in the <u>http://clok.uclan.ac.uk/policies/</u>

#### LIVint-19-01685

**TITLE:** Incidence, prevalence, and mortality of autoimmune hepatitis in England 1997-2015. A population-based cohort study.

#### **AUTHORS AND AFFILIATIONS**

Lisbet Grønbæk, MD, PhD<sup>1, 2, 3</sup>: groenbaek.lisbet@gmail.com

Harmony Otete, MPH, PhD, Assistant Professor <sup>3, 4</sup>: HOtete@uclan.ac.uk

\*Lu Ban, BSc, MSc, PhD, Senior Research Fellow <sup>5, 6</sup>: lu.ban@evidera.com

Colin Crooks, MD, PhD, Clinical Associate Professor <sup>5, 6</sup>: Colin.Crooks@nottingham.ac.uk

Timothy Card, MD, PhD, Clinical Associate Professor<sup>3,5</sup>: Tim.Card@nottingham.ac.uk

Peter Jepsen, MD, PhD, Clinical Associate Professor<sup>1, 2, 3</sup>: pj@clin.au.dk

Joe West, Professor of Epidemiology<sup>3,5</sup>: Joe.West@nottingham.ac.uk

- Department of Hepatology and Gastroenterology, Aarhus University Hospital, Palle Juul-Jensens Boulevard 99, 8200 Aarhus N, Denmark.
- Department of Clinical Epidemiology, Aarhus University Hospital, Olof Palmes Allé 43-45, 8200 Aarhus N, Denmark.
- Division of Epidemiology and Public Health, School of Medicine, University of Nottingham, Nottingham NG7 2UH, United Kingdom.
- School of Medicine and Dentistry, University of Central Lancashire, Preston, Lancashire PR1 2HE, United Kingdom.

- National Institute for Health Research (NIHR) Nottingham Biomedical Research Centre, the Nottingham University Hospitals NHS Trust and University of Nottingham, Nottingham NG7 2UH, United Kingdom.
- Nottingham Digestive Diseases Centre, School of Medicine, University of Nottingham, Nottingham, NG7 2UH, United Kingdom.

\* Footnote: Dr Lu Ban's present affiliation with address is Evidera by PPD, The Ark, 201 Talgarth Rd, London, W6 8BJ, United Kingdom

# **CORRESPONDING AUTHOR**

Lisbet Grønbæk

Department of Hepatology and Gastroenterology, Aarhus University Hospital

Palle Juul-Jensens Boulevard 99, 8200 Aarhus N, Denmark

Telephone: +45 78453813; Fax: +45 78453897; Email address: groenbaek.lisbet@gmail.com

## Word count for main body of manuscript: 3,944 words

Figures and tables: 3 figures and 4 tables

# **ABBREVIATIONS**

Autoimmune Hepatitis (AIH); United Kingdom (UK); Clinical Practice Research Datalink (CPRD); Hospital Episode Statistics (HES); International Classification of Diseases 10<sup>th</sup> edition (ICD-10); Office of Population Censuses and Surveys (OPCS); Index of Multiple Deprivation (IMD); Office of National Statistics' (ONS); Hepatitis B Virus (HBV); Hepatitis C Virus (HCV); Primary Sclerosing Cholangitis (PSC); Primary Biliary Cholangitis (PBC); Hepatocellular Carcinoma (HCC); Confidence Interval (CI); Non-alcoholic fatty liver disease (NAFLD); Non-alcoholic steatohepatitis (NASH).

# **CONFLICTS OF INTEREST**

The authors who have taken part in this study declare that they have no conflict of interest with respect to this manuscript.

#### FINANCIAL SUPPORT

Lisbet Grønbæk received funding from the Danish foundation of 17.12.1981 and the Danish Foundation of A.P. Møller. The funding sources were not involved in the conduct of the research or preparation of the article.

## ACKNOWLEDGEMENTS

This study is based in part on data from the Clinical Practice Research Datalink (CPRD) obtained under the University of Nottingham licence from the United Kingdom Medicines and Healthcare Products Regulatory Agency. The anonymised data is provided by patients and collected by the NHS as part of their care and support. The interpretation and conclusion contained in this study are those of the authors alone. Hospital Episode Statistics Data copyright ® (2015), reused with the permission of the Health & Social Care Information Centre, all rights reserved.

#### ABSTRACT

**Background & Aims:** There are few population-based studies of the incidence and mortality of autoimmune hepatitis. The burden of the disease and how it has changed over time have not been fully explored. We conducted a population-based cohort study on the incidence and mortality of autoimmune hepatitis in England, 1997–2015.

**Methods:** From the Clinical Practice Research Datalink we included 882 patients diagnosed with autoimmune hepatitis in England, 1997-2015. The patients were followed through 2015, and we calculated the sex- and age- standardised incidence and prevalence of autoimmune hepatitis. We examined variation in incidence by sex, age, calendar year, geographical region, and socioeconomic status, and incidence rate ratios were calculated with Poisson regression. We calculated all-cause and cause-specific mortality.

**Results:** The overall standardised incidence rate of autoimmune hepatitis was 2.08 (95% confidence interval 1.94-2.22) per 100,000 population per year, higher in women, higher in older age, and independent of region and socioeconomic status. From 1997 to 2015 the incidence doubled from 1.27 (95% confidence interval 0.51-2.02) to 2.56 (95% confidence interval 1.79-3.33) per 100,000 population per year. The 10-year cumulative all-cause mortality was 31.9% (95% confidence interval 27.6-36.5), and the 10-year cumulative liver-related mortality, including hepatocellular carcinoma was ~ 10.5%.

**Conclusions:** This population-based study showed that the incidence of autoimmune hepatitis doubled over an eighteen-year period. The incidence was particularly high in older women and was similar across all regions of England and independent of socioeconomic status. Patients with autoimmune hepatitis had a high mortality.

**KEYWORDS:** epidemiology; registry; cohort study; hepatitis, autoimmune; incidence; prevalence; prognosis.

4

# Lay summary

Autoimmune hepatitis is a chronic inflammatory liver disease. We studied the English population and found that the incidence of autoimmune hepatitis doubled from 1997 to 2015. The resulting prevalence was 19 per 100,000 population, i.e. more than 10,000 English people live with AIH in 2015. The patients with autoimmune hepatitis had a high mortality with a 10-year mortality of 32%.

#### **INTRODUCTION**

Autoimmune hepatitis (AIH) is a female-predominated chronic inflammatory liver disease. The aetiology is complex and likely involves interactions between genetic and environmental factors.<sup>1,2</sup> The incidence rate of AIH from various regions of the world ranges from 0.4 to 3.0 per 100,000 population per year.<sup>3-7</sup> Some authors have reported a stable incidence over time,<sup>5,8-10</sup> others an increasing incidence.<sup>3,6,7,11-14</sup> Apart from sex, age, and calendar time, socioeconomic status and region of residence may affect the likelihood of being diagnosed with AIH.<sup>15,16</sup> Evaluation of the variation in AIH incidence over time and place may provide novel insight into the risk factors for AIH. For this purpose, population-based studies over long periods are crucial.

AIH is regarded a treatable disease with effective immunosuppressive medical therapy.<sup>17</sup> Even so, the reported 10-year cumulative mortality for patients with AIH ranges widely from 4% to 74%. Some previous studies concluded that patients with AIH have a higher mortality than the general population,<sup>5,7,14,18-23</sup> others that patients with AIH have a similar mortality to that of the general population.<sup>24-29</sup>

We conducted a population-based study of AIH over nearly two decades and across all regions of England. We aimed to report the incidence and prevalence of AIH and to determine trends and variation in incidence and prevalence by sex, age, calendar year, socioeconomic status, and geographical region. We also aimed to report data on mortality, causes of death, and liver transplantation for patients with AIH.

#### MATERIALS AND METHODS

This is a population-based cohort study of English patients registered with a diagnosis of AIH in primary or secondary care between 1997 and 2015. <u>The study is population-based using routine healthcare records, and no patient enrolment was involved.</u> The study was approved by the

Independent Scientific Advisory Committee for MHRA Database Research, United Kingdom (UK) (protocol 17\_110RA).

#### **Data sources**

We extracted data from the Clinical Practice Research Datalink (CPRD) that contains records collected from general practice in the UK since 1987.<sup>30</sup> These records include sex, dates of birth and death, dates of diagnoses defined by Read codes, biochemical tests, and medical prescriptions. Read codes are recorded in primary care on the basis of clinical consultations and other resources such as hospital discharge notes and discharge diagnoses codes. General practices usually upload data to the CPRD on a monthly basis, and a dataset available for research use is concurrently generated on that basis. Data quality is assessed by the CPRD against criteria defining patient-level 'acceptability' and practice-level 'up-to-standard' time quality. The CPRD is representative of the overall UK population in terms of age, sex, and ethnicity.<sup>30</sup> 57% of the CPRD practices have consented to linkage to other data sources. Since 1997, individual-level data has been linked with data from the Hospital Episode Statistics (HES) database. This database includes dates of admission and discharge, discharge diagnoses coded according to the International Classification of Diseases 10th edition (ICD-10), and procedure codes according to the Office of Population Censuses and Surveys (OPCS) classification. The CPRD is also linked to deprivation data defined by quintiles of Index of Multiple Deprivation (IMD) at a general practice-level and at a patient-level.<sup>31</sup> For the current study we used practice-level data because the denominator is the whole of the CPRD population, and access to deprivation data for each individual was not permitted. A patient is assigned an IMD score based on the average socioeconomic status in the area (each containing around 150 dwellings), in which the general practice is situated. Deprivation data in such an area have been found to be a reasonable proxy for the level of deprivation of the entire general practice population.<sup>32</sup> Since 1998, linkage to the Office of National Statistics' (ONS) death registry has been available. The registry includes information on causes of death based on information from death certificates. The causes of death are

classified according to the ICD-10, and for each patient one *underlying* cause of death is defined by World Health Organisation standardised criteria.<sup>33</sup>

#### **Study population**

We included all permanently registered patients in the CPRD, who were acceptable for research and eligible for linkage between 1 April 1997 and 31 December 2015. Cases of AIH were defined by CPRD registration of a Read code compatible with AIH or by HES registration of an ICD-10 diagnosis of AIH between 1 April 1997 and 31 December 2015 (Read codes: J63B.00 autoimmune hepatitis, J614000 chronic persistent hepatitis, J614111 autoimmune chronic active hepatitis, or J614100 chronic active hepatitis; ICD-10 codes: K73.2 chronic active hepatitis or K75.4 autoimmune hepatitis). We excluded patients who at any time during follow-up were registered with a <u>Read code or an ICD-10</u> code compatible with alcoholic fibrosis or cirrhosis; viral hepatitis, i.e. hepatitis B virus (HBV) or hepatitis C virus (HCV), a positive test result for HBV or HCV, or an antiviral treatment for HBV or HCV; or another autoimmune liver disease, i.e. primary sclerosing cholangitis (PSC) or primary biliary cholangitis (PBC) (Supplementary Table 1 shows codes for exclusion).

A patient was identified as incident, if 1) the first Read code or ICD code for AIH occurred between 1 April 1997 and 31 December 2015; 2) the patient had been registered at the same general practice for at least 12 months before the diagnosis; and 3) data from that general practice were up-to-standard. The 12-month time gap after registration ensured that prevalent AIH patients were not misclassified as incident ones.<sup>34</sup> Patients with an AIH diagnosis date concurrent with a date of death were excluded. All other patients with a code for AIH occurring between 1 April 1997 and 31 December 2015 were considered prevalent AIH cases. Incident AIH patients were classified as having or not having cirrhosis at the time of AIH diagnosis. Cirrhosis was defined by registration of

a code compatible with cirrhosis or a cirrhosis complication before or within the 6-month period after the time of AIH diagnosis (Supplementary Table 2 shows codes defining cirrhosis).

#### **Death and transplantation**

Information on date of death and underlying cause of death was obtained from the ONS death registry. Underlying causes of death were categorised as follows: Liver-related (including cirrhosis and cirrhosis complications other than infections and hepatocellular carcinoma [HCC]); HCC; any other cancer; infection; cardiovascular disease (including cerebrovascular disease and excluding infections); respiratory cause (excluding infections); and other causes of death. Information on liver transplantations was obtained from the HES database (Supplementary Table 3 shows codes for causes of death and liver transplantation).

#### Statistical analyses

#### Incidence and prevalence

The person-time at risk for developing AIH for an individual began after 1 January 1997, when the patient had been registered at a general practice for at least 12 months, and after the data were up-tostandard. The person-time at risk ended at the time of AIH diagnosis, cessation of registration at the general practice, the last data collection by the CPRD, death, or end of study on 31 December 2015 (whichever came first). We calculated the overall incidence rate of AIH as the number of incident AIH patients divided by the total person-time at risk. The incidence rates were stratified by sex, age group, calendar year, practice-level IMD, and geographical region. We calculated the point prevalence of AIH on 31 December 2015 as the number of AIH patients diagnosed before and still alive on that date divided by the total number of included CPRD patients who were alive on that date. We used direct standardization to standardise incidence and prevalence rates to the European standard population's sex and age distribution in 2013.<sup>35</sup> We evaluated the time trend in AIH incidence using calendar year as a continuous variable in a Poisson regression model fitted to estimate incidence rate ratios of AIH, adjusted for sex, age group, IMD, and geographical region. The likelihood ratio test was used to test for interactions between calendar year, sex, and age. A Poisson regression model for each sex was then applied to estimate adjusted incidence rate ratios. We extracted the percentage-wise change in AIH incidence per calendar year.

#### Sensitivity analyses

We repeated calculations of AIH incidence rates *including* AIH patients with potential overlap diagnoses with PSC or PBC, i.e. patients who at any time during follow-up were registered with a code compatible with PSC or PBC.

# Mortality and transplantation

The person-time for risk of death began at the time of AIH diagnosis and ended at death, cessation of registration at the general practice, the last data collection by the CPRD, or end of study on 31 December 2015 (whichever came first). For our cohort of incident AIH patients, we firstly calculated the all-cause mortality rate as the number of deaths divided by the total person-time at risk of death, and we calculated the 5- and 10-year cumulative mortality based on the Kaplan Meier estimator. The rates were stratified by sex, age groups, IMD, and cirrhosis. Secondly, we calculated cause-specific mortality rates as the number of deaths from a specific underlying cause divided by the total person-time at risk of death, and we calculated the 5- and 10-year cumulative mortality from each cause based on the cumulative incidence function allowing for the competing risk from other causes of death.<sup>36</sup> Thirdly, we calculated the 5- and 10-year cumulative incidence of liver transplantation with death as a competing risk, and we calculated the 5- and 10- year cumulative incidence of death without transplantation, with transplantation as the competing risk. We used Stata version 14.2 for all analyses.

#### RESULTS

#### **Incidence and prevalence**

We identified 1,008 patients diagnosed with AIH between 1997 and 2015, but 126 were excluded from our primary analysis: 31 with a diagnosis of alcoholic liver disease or with a diagnosis date concurrent with their date of death (<5 of whom on the basis of the latter); 33 with HBC or HCV; and 62 with a diagnosis of PSC or PBC.

Our final cohort included 882 incident patients with AIH from 10 regions of England; 662 (75.1%) were women (median age at diagnosis was 60 years [interquartile range 49-71]). The remaining 220 (24.9%) were men (median age at diagnosis was 54 years [interquartile range 39-68]). There were no missing data on sex, age, geographical region, or IMD. In the 882 patients with AIH, 156 (17.7%) had cirrhosis at the time of diagnosis, and a total of 26.0% had developed cirrhosis at the end of follow-up.

For both men and women, the overall incidence rate peaked around 70 years of age (Figure 1). For the 1997-2015 period, the standardised incidence rate was 2.08 (95% confidence interval [CI] 1.94-2.22) per 100,000 population per year with a doubling in incidence from 1997 to 2015 (Table 1 and Figure 2). The standardised incidence rate was higher in women than in men. The standardised point prevalence on 31 December 2015 was 19.24 (95% CI 18.08-20.41) per 100,000 population.

The incidence increased between 1997 and 2015 with an average increment per calendar year of 3.1% (95% CI 1.5 to 4.8) for women and 2.5% (95% CI 0.0-5.3) for men. In the adjusted model the incidence was similar regardless of IMD and geographical region (Table 2). We found a three-way interaction between sex, age, and calendar year (likelihood ratio test, p=0.003), in that the incidence increment over calendar time was mainly apparent in women of older age.

The sensitivity analysis including AIH patients with potential overlap diagnoses with PSC or PBC comprised 944 patients for an overall standardised incidence rate of 2.23 (95% CI 2.09-2.37) per 100,000 population per year. The inclusion of these patients did not change the incidence patterns.

#### Mortality and transplantation

Of the 882 patients with AIH, 191 (21.7%) died during a total of 4,216 years of follow-up. The mortality rate was particularly high in the first year after AIH diagnosis compared with the rate thereafter: 10.00 (95% CI 8.00-12.50) vs. 3.31 (95% CI 2.75-3.98) per 100 patients per year. The 10-year all-cause cumulative mortality was 31.9% (95% CI 27.6-36.5). The risk of death was higher for those diagnosed with AIH in older age, in the deprived, and in those with cirrhosis (Table 3). Information on underlying cause of death was missing for less than five persons. Liver-related causes of death and death from cancers other than HCC dominated (Figure 3), and the mortality rates were especially high in the first year after diagnosis (Table 4).

Thirteen patients with AIH, of whom five had cirrhosis at the time of diagnosis, had a livertransplantation during the follow-up with a 5-year cumulative incidence of transplantation of 1.3% (95% CI 0.7-2.4) and a 10-year cumulative incidence of 2.2% (95% CI 1.1-3.7). The 5- and 10-year cumulative incidence of death without transplantation was 19.3% (95% CI 16.4-22.4) and 31.4% (95% CI 27.0-35.8), respectively.

#### DISCUSSION

This population-based study from England showed the highest incidence of AIH among women in their seventies. There was no association with region of residence or socioeconomic status. We found that the incidence of AIH doubled from 1997 to 2015, resulting in a prevalence of 19 per 100,000 population. The 10-year all-cause cumulative mortality was 32% with liver-related cause of

death being predominant. The mortality rate was particularly high during the first year after diagnosis.

Our study is one of the largest population-based studies on AIH incidence spanning more than a decade. Including all regions of England, the study cohort is representative of the entire English population.<sup>30</sup> The completeness of the recorded diagnoses may be a concern. With the routine record-based approach we cannot be certain that every patient with AIH was captured. However, we included patients identified in both primary, secondary, and tertiary care. Our incidence and prevalence estimates are in accordance with other studies using registry diagnoses for AIH patient identification<sup>7,14</sup> as well as in accordance with studies using medical notes, biochemistry, and histology for patient identification.<sup>3,5,6,8-13</sup> This indicates that our case ascertainment is likely to be reasonably complete.

The validity of the routine data diagnoses may also be a concern, as although the validity of diagnostic coding in general practice is high, <sup>37,38</sup> we did not have liver biopsy data or the data to calculate an AIH score to validate the AIH diagnoses<sup>39,40</sup>. However, our current study did use all available data within the routine records combining Read codes we previously identified in primary care<sup>41</sup> with case ascertainment from secondary and tertiary care using selected ICD-10 codes. In the previous study<sup>41</sup> the identified Read codes applied to patients discharged with a valid AIH diagnosis from tertiary care, i.e. patients who had probable or definite AIH according to the International Autoimmune Hepatitis Group simplified score<sup>40</sup> or a liver biopsy compatible with AIH, and who had been given an AIH diagnosis by a hepatologist consultant. We excluded possibly misclassified cases of alcoholic and other autoimmune liver diseases. As for PSC, the ICD-10 code may be unspecific, and therefore we might have excluded cholangitis patients with other aetiology than PSC. We will stress that the exclusion was primarily based on Read codes registered in the CPRD data, and those codes were merely supplemented with ICD-10 codes from the HES. Therefore, it is unlikely that we

greatly under-ascertained the prevalence and incidence of AIH. We did not exclude patients with non-alcoholic fatty liver disease (NAFLD) and non-alcoholic steatohepatitis (NASH). Although this might be a limitation, the awareness of NAFLD and NASH and the completeness of registered diagnoses in the CPRD and the HES have probably changed over time. Hence, we were not confident to use diagnoses codes for these conditions as exclusion criteria. Furthermore, for the patients with AIH any such misdiagnosis is likely to be less of a limitation as these conditions would most likely have been ruled out based on presence of the metabolic syndrome, liver histology, and/or autoantibody profile during the diagnostic workup for AIH. Therefore, it is also unlikely that we greatly over-ascertained the prevalence and incidence of AIH. With the exclusion of patients with diagnoses of the other chronic liver diseases that may have higher associated mortality, our mortality estimates are too unlikely to have been overestimated.

The overall standardised incidence rate of 2.1 per 100,000 population per year in our study is similar to the incidence rates reported in our previous Danish study<sup>14</sup>, a New Zealand study,<sup>8</sup> and a recent Icelandic study,<sup>6</sup> all population-based. We confirmed the well-known female predominance of AIH. In line with previous studies from the UK<sup>9,42-44</sup> and studies from others parts of the world<sup>3,6-8,10,11,13,14,19,45-48</sup> we found an incidence peak in older age. We could not confirm a peak in younger age that was found in some of the studies,<sup>10,14,19,48</sup> and that might relate to risk factor exposures in younger age in some countries. Yet, we cannot rule out that our definition of incident cases may have biased against detecting cases in the younger subset of the CPRD population, who are more mobile and less likely to be permanently registered with the same CPRD general practice.

In the patients with AIH, we found a lower proportion of cirrhosis than in most other studies. However, these studies were from tertiary centres that might have been biased by only including more severe cases of AIH.<sup>10,11,29,47,49</sup> Our study, on the other hand, also includes patients from primary and secondary care and might therefore be able to identify patients diagnosed at an earlier stage before developing cirrhosis. Two population-based studies found an even lower proportion of cirrhosis at the time of diagnosis than we did,<sup>6,13</sup> and compared with the Danish study, we found a similar final proportion of cirrhosis.<sup>14</sup>

In line with a population-based case-control study,<sup>16</sup> we found no association between socioeconomic status and the incidence of AIH. One review found an association between socioeconomic status (high human development index) on a country level and high AIH incidence,<sup>15</sup> but such a finding does not necessarily reflect an association on an individual level.<sup>50</sup> We did not have access to patient-level socioeconomic status, either, but deprivation data was obtained from smaller areas with presumably smaller variation compared with variation within whole countries.

With a doubling of AIH incidence from 1997 to 2015, we confirmed the findings of our previous Danish study<sup>14</sup> and other population-based studies.<sup>67,13</sup> Some authors have reported a stable incidence over time,<sup>5,8-10</sup> yet others an increasing incidence, but arguing that the increase was a result of increased diagnostic activity.<sup>3,11,12</sup> The current study and a previous English study<sup>51</sup> show higher incidence rates than older UK studies from the 1970s and 1990s.<sup>9,44</sup> This may partly be explained by an increased diagnostic ascertainment over time,<sup>3,11,12</sup> but in preliminary data analyses, 2005-2015, there was an overall decline in the testing rates of liver function tests within the whole CPRD dataset (personal communication, Dr Lu Ban). Moreover, if increased diagnostic ascertainment was the only explanation, the incidence would eventually plateau, as the clinical diagnosis rate approached the underlying, true population rate. Similar incidence time trends were reported for many other autoimmune diseases, such as PSC, rheumatoid arthritis, inflammatory bowel disease, and diabetes,<sup>52,53</sup> and therefore there might be an increasing overall awareness of autoimmune diseases such as polymyalgia rheumatica has remained relatively stable over time,<sup>54</sup> for coeliac disease it has increased and then stabilized,<sup>55</sup> and for dermatitis herpetiformis it has declined.<sup>56</sup> We cannot exclude

the possibility that our results partly reflect an increasing incidence of AIH *diagnoses* rather than an increasing incidence of the *disease* AIH, and any such concern would apply to the majority of reported disease incidence estimates. Our study, the Danish,<sup>14</sup> and the Dutch study<sup>13</sup> were large and spanned a long period with sustained incidence increases, and we believe that more people actually developed AIH. Genetic susceptibility to AIH is unlikely to have varied over time to an extent that would have caused a doubling in incidence over 18 years. Therefore, environmental factors may be the more likely explanation, with risk factors either accumulating over time, or protective factors being lost.<sup>57</sup> A recent population-based study, 2006-2015, from Iceland suggested that drug-induced AIH, mainly due to biological drugs, could explain the incidence increase.<sup>6</sup> In the original AIH diagnostic score, revised in 1999,<sup>39</sup> the use of hepatotoxic drugs would count against a diagnosis of AIH, whereas in the simplified AIH score, 2008,<sup>40</sup> that criterion was not included. For our study period, both scoring systems could have been used, and an increasing use of hepatotoxic drugs such as biological drugs may have contributed to our findings.

In line with the Danish study<sup>14</sup> and a recent study from Finland<sup>7</sup> the incidence increase was more distinct in women than in men, particularly in older women. These findings might be attributable to a lower threshold for autoimmune trigger mechanisms in women due to female hormones, whereas the interpretation of the interaction with older age remains unclear. Future studies of risk factors for AIH will need to make separate analyses for men and women and for the young and the old.

In line with most previous studies,<sup>5,7,14,18-23</sup> our findings support the notion that patients with AIH have a high mortality. <u>We were unable to include details on hospital care or medical management of the patients, as CPRD data, even with linkage with HES, cannot reliably determine such details.</u> We found a 10-year mortality risk of 32% that was higher than in the other studies, and this is likely to be partially explained by a higher median age of our study cohort. Lower age undoubtedly also contributed to the findings of mortality risks as low as 4% from other studies, but the selection of

patients with AIH in those studies could also be of significance: one study solely included patients with type 1 AIH;<sup>26</sup> other studies were single-centre studies and may consequently have included a selected group of patients with a favourable prognosis;<sup>24,25,29</sup> and two studies included Japanese patients, who often present with less severe disease.<sup>27,28</sup> Furthermore, one of the Japanese studies<sup>28</sup> evaluated survival only in patients who were followed more than 12 months after AIH diagnosis. These patients would have had to survive 12 months after diagnosis to be included in the analysis, and this would have biased the survival curve upwardly.<sup>58</sup> Likewise, one English study<sup>29</sup> only included patients who had a diagnostic liver biopsy *and* a further biopsy while on medical treatment for AIH, and this could also have biased the survival curve.<sup>58</sup> Another English study included 245 patients who fulfilled the diagnostic criteria for AIH and found a higher mortality than in the general population.<sup>23</sup> The results from our current larger population-based study suggest that AIH in England is still a dangerous disease, and in line with other studies, <sup>5,14,18,19,21,22,29</sup> the mortality was higher in those diagnosed in older age and in those with cirrhosis The excess mortality rate in those with a lower affluence corresponds to the findings from a study on deprivation and associated mortality in England and Wales.<sup>59</sup>

One single centre study on mortality in patients with AIH<sup>26,27</sup> found that liver-related death accounted for as little as 0%,<sup>60</sup> others found it to account for as much as 50%,<sup>26,27</sup> but these studies were small. Larger population-based studies with long-term follow-up<sup>5,7,14,21,22</sup> found a frequency of liver-related death, including HCC, of 32-49% and death from other cancers of 15-25%; and our findings are comparable to theirs. Extending the results from the other studies, we quantified the cause-specific cumulative risks of death in the patients with AIH, and we found a high long-term mortality risk that was mainly liver- and cancer-related. The all-cause mortality rate and the rate from liver-related causes and other cancers were particularly high in the first year following AIH diagnosis. The current results substantiate the finding of a pronounced excess mortality in the first year after diagnosis from our previous Danish study.<sup>14</sup> In line with some,<sup>7,14,20,21</sup> but not all,<sup>5,22</sup>

population-based studies, only 1.5% of our English study cohort had a liver transplantation. This proportion is slightly higher than expected, based on the 2017/2018 National Health Service annual report on liver transplantations in the UK,<sup>61</sup> and even more patients with AIH might benefit from transplantation.

In conclusion, we found a high incidence of AIH in England, particularly in women in their seventies. The incidence of AIH in England is increasing, and studies to identify risk factors are increasingly important. We found a high mortality in English patients with AIH, and the mortality rate from liver- and cancer-related causes were especially high in the first year after AIH diagnosis.

#### REFERENCES

1. Manns MP, Lohse AW, Vergani D. Autoimmune hepatitis - Update 2015. *Journal of hepatology* 2015; 62(1S): S100-S11.

2. Grønbæk L, Vilstrup H, Pedersen L, Christensen K, Jepsen P. Family occurrence of autoimmune hepatitis: A Danish nationwide registry-based cohort study. *J Hepatol* 2018; 69(4): 873-7.

3. Yoshizawa K, Joshita S, Matsumoto A, et al. Incidence and prevalence of autoimmune hepatitis in the Ueda area, Japan. *Hepatology research : the official journal of the Japan Society of Hepatology* 2015.

Jepsen P, Grønbæk L, Vilstrup H. Worldwide Incidence of Autoimmune Liver Disease. *Dig Dis* 2015; 33 Suppl 2: 2-12.

5. Danielsson Borssen A, Marschall HU, Bergquist A, et al. Epidemiology and causes of death in a Swedish cohort of patients with autoimmune hepatitis. *Scand J Gastroenterol* 2017; 52(9): 1022-8.

6. Valgeirsson KB, Hreinsson JP, Bjornsson ES. Increased incidence of autoimmune hepatitis is associated with wider use of biological drugs. *Liver Int* 2019.

7. Puustinen L, Barner-Rasmussen N, Pukkala E, Farkkila M. Incidence, prevalence, and causes of death of patients with autoimmune hepatitis: A nationwide register-based cohort study in Finland. *Dig Liver Dis* 2019; 51(9): 1294-9.

 Ngu JH, Bechly K, Chapman BA, et al. Population-based epidemiology study of autoimmune hepatitis: A disease of older women? *Journal of gastroenterology and hepatology* 2010; 25(10): 1681-6. 9. Tanner AR, Dellipiani AW. Chronic active hepatitis: a sixteen year survey at a district general hospital. *Postgrad Med J* 1989; 65(768): 725-8.

10. Werner M, Prytz H, Ohlsson B, et al. Epidemiology and the initial presentation of autoimmune hepatitis in Sweden: a nationwide study. *Scandinavian Journal of Gastroenterology* 2008; 43(10): 1232-40.

11. Boberg KM, Aadland E, Jahnsen J, Raknerud N, Stiris M, Bell H. Incidence and prevalence of primary biliary cirrhosis, primary sclerosing cholangitis, and autoimmune hepatitis in a Norwegian population. *Scandinavian Journal of Gastroenterology* 1998; 33(1): 99-103.

12. Primo J, Merino C, Fernandez J, Moles JR, Llorca P, Hinojosa J. Incidence and prevalence of autoimmune hepatitis in the area of the Hospital de Sagunto (Spain). *Gastroenterologia y hepatologia* 2004; 27(4): 239-43.

13. van Gerven NM, Verwer BJ, Witte BI, et al. Epidemiology and clinical characteristics of autoimmune hepatitis in the Netherlands. *Scand J Gastroenterol* 2014; 49(10): 1245-54.

14. Grønbæk L, Vilstrup H, Jepsen P. Autoimmune hepatitis in Denmark: incidence, prevalence, prognosis, and causes of death. A nationwide registry-based cohort study. *J Hepatol* 2014; 60(3): 612-7.

15. Pan HY, Dai YN, Zheng JN, et al. National incidence of autoimmune liver diseases and its relationship with the human development index. *Oncotarget* 2016; 7(29): 46273-82.

16. Ngu JH, Gearry RB, Frampton CM, Stedman CA. Autoimmune hepatitis: the role of environmental risk factors: a population-based study. *Hepatol Int* 2013; 7(3): 869-75.

17. Cook GC, Mulligan R, Sherlock S. Controlled prospective trial of corticosteroid therapy in active chronic hepatitis. *The Quarterly journal of medicine* 1971; 40(158): 159-85.

18. Feld JJ, Dinh H, Arenovich T, Marcus VA, Wanless IR, Heathcote EJ. Autoimmune hepatitis:
effect of symptoms and cirrhosis on natural history and outcome. *Hepatology (Baltimore, Md)* 2005;
42(1): 53-62.

19. Landeira G, Morise S, Fassio E, et al. Effect of cirrhosis at baseline on the outcome of type 1 autoimmune hepatitis. *Annals of hepatology* 2012; 11(1): 100-6.

20. Ngu JH, Gearry RB, Frampton CM, Stedman CA. Mortality and the risk of malignancy in autoimmune liver diseases: a population-based study in Canterbury, New Zealand. *Hepatology (Baltimore, Md)* 2012; 55(2): 522-9.

21. van den Brand FF, van der Veen KS, de Boer YS, et al. Increased Mortality Among Patients With vs Without Cirrhosis and Autoimmune Hepatitis. *Clin Gastroenterol Hepatol* 2018.

22. Werner M, Wallerstedt S, Lindgren S, et al. Characteristics and long-term outcome of patients with autoimmune hepatitis related to the initial treatment response. *Scandinavian Journal of Gastroenterology* 2010; 45(4): 457-67.

23. Hoeroldt B, McFarlane E, Dube A, et al. Long-term outcomes of patients with autoimmune hepatitis managed at a nontransplant center. *Gastroenterology* 2011; 140(7): 1980-9.

24. Kirstein MM, Metzler F, Geiger E, et al. Prediction of short- and long-term outcome in patients with autoimmune hepatitis. *Hepatology (Baltimore, Md)* 2015.

25. Malekzadeh Z, Haghazali S, Sepanlou SG, et al. Clinical features and long term outcome of 102 treated autoimmune hepatitis patients. *Hepat Mon* 2012; 12(2): 92-9.

26. Roberts SK, Therneau TM, Czaja AJ. Prognosis of histological cirrhosis in type 1 autoimmune hepatitis. *Gastroenterology* 1996; 110(3): 848-57.

27. Migita K, Watanabe Y, Jiuchi Y, et al. Hepatocellular carcinoma and survival in patients with autoimmune hepatitis (Japanese National Hospital Organization-autoimmune hepatitis prospective study). *Liver international : official journal of the International Association for the Study of the Liver* 2012; 32(5): 837-44.

28. Yoshizawa K, Matsumoto A, Ichijo T, et al. Long-term outcome of Japanese patients with type 1 autoimmune hepatitis. *Hepatology (Baltimore, Md)* 2012; 56(2): 668-76.

29. Dhaliwal HK, Hoeroldt BS, Dube AK, et al. Long-Term Prognostic Significance of Persisting Histological Activity Despite Biochemical Remission in Autoimmune Hepatitis. *Am J Gastroenterol* 2015; 110(7): 993-9.

30. Herrett E, Gallagher AM, Bhaskaran K, et al. Data Resource Profile: Clinical Practice Research Datalink (CPRD). *Int J Epidemiol* 2015; 44(3): 827-36.

31. Ministry of Housing CLG, UK. https://www.gov.uk/government/statistics/english-indices-ofdeprivation-2015. 2015 (accessed 30 September 2015).

32. Strong M, Maheswaran R, Pearson T. A comparison of methods for calculating general practice level socioeconomic deprivation. *Int J Health Geogr* 2006; 5: 29.

33. Office for National Statistics U.

https://www.ons.gov.uk/peoplepopulationandcommunity/birthsdeathsandmarriages/deaths/methodol ogies/userguidetomortalitystatisticsjuly2017. 2017.

34. Lewis JD, Bilker WB, Weinstein RB, Strom BL. The relationship between time since registration and measured incidence rates in the General Practice Research Database. *Pharmacoepidemiol Drug Saf* 2005; 14(7): 443-51.

35. Eurostat. https://ec.europa.eu/eurostat/documents/3859598/5926869/KS-RA-13-028-EN.PDF/e713fa79-1add-44e8-b23d-5e8fa09b3f8f

36. Jepsen P, Vilstrup H, Andersen PK. The clinical course of cirrhosis: The importance of multistate models and competing risks analysis. *Hepatology* 2015; 62(1): 292-302.

37. Khan NF, Harrison SE, Rose PW. Validity of diagnostic coding within the General Practice Research Database: a systematic review. *Br J Gen Pract* 2010; 60(572): e128-36.

38. Herrett E, Thomas SL, Schoonen WM, Smeeth L, Hall AJ. Validation and validity of diagnoses in the General Practice Research Database: a systematic review. *Br J Clin Pharmacol* 2010; 69(1): 4-14.

39. Alvarez F, Berg PA, Bianchi FB, et al. International Autoimmune Hepatitis Group Report: review of criteria for diagnosis of autoimmune hepatitis. *Journal of hepatology* 1999; 31(5): 929-38.

40. Hennes EM, Zeniya M, Czaja AJ, et al. Simplified criteria for the diagnosis of autoimmune hepatitis. *Hepatology* 2008; 48(1): 169-76.

41. Varyani F, Card T, Kaye P, Aithal GP, West J. The communication of a secondary care diagnosis of autoimmune hepatitis to primary care practitioners: a population-based study. *BMC Health Serv Res* 2013; 13: 161.

42. Al-Chalabi T, Boccato S, Portmann BC, McFarlane IG, Heneghan MA. Autoimmune hepatitis (AIH) in the elderly: a systematic retrospective analysis of a large group of consecutive patients with definite AIH followed at a tertiary referral centre. *J Hepatol* 2006; 45(4): 575-83.

43. Parker DR, Kingham JG. Type I autoimmune hepatitis is primarily a disease of later life. *QJM* 1997; 90(4): 289-96.

44. Hodges JR, Millward-Sadler GH, Wright R. Chronic active hepatitis: the spectrum of disease. *Lancet* 1982; 1(8271): 550-2.

45. Schramm C, Kanzler S, zum Buschenfelde KH, Galle PR, Lohse AW. Autoimmune hepatitis in the elderly. *Am J Gastroenterol* 2001; 96(5): 1587-91.

46. Ritland S. The incidence of chronic active hepatitis in Norway. A retrospective study. *Scand J Gastroenterol Suppl* 1985; 107: 58-60.

47. Lee YM, Teo EK, Ng TM, Khor C, Fock KM. Autoimmune hepatitis in Singapore: a rare syndrome affecting middle-aged women. *Journal of gastroenterology and hepatology* 2001; 16(12): 1384-9.

48. Muratori P, Granito A, Quarneti C, et al. Autoimmune hepatitis in Italy: the Bologna experience. *J Hepatol* 2009; 50(6): 1210-8.

49. Ngu JH, Gearry RB, Frampton CM, Stedman CA. Predictors of poor outcome in patients with autoimmune hepatitis: A population-based study. *Hepatology (Baltimore, Md)* 2013.

50. Wakefield J. Ecologic studies revisited. Annu Rev Public Health 2008; 29: 75-90.

51. Whalley S, Puvanachandra P, Desai A, Kennedy H. Hepatology outpatient service provision in secondary care: a study of liver disease incidence and resource costs. *Clin Med (Lond)* 2007; 7(2): 119-24.

52. Lerner A, Jeremias P, Matthias T. The World Incidence and Prevalence of Autoimmune Diseases is Increasing. *International Journal of Celiac Disease* 2015; 3(4): 151-5.

53. Card TR, Solaymani-Dodaran M, West J. Incidence and mortality of primary sclerosing cholangitis in the UK: a population-based cohort study. *J Hepatol* 2008; 48(6): 939-44.

54. Partington RJ, Muller S, Helliwell T, Mallen CD, Abdul Sultan A. Incidence, prevalence and treatment burden of polymyalgia rheumatica in the UK over two decades: a population-based study. *Ann Rheum Dis* 2018; 77(12): 1750-6.

55. West J, Otete H, Sultan AA, Crooks CJ. Changes in Testing for and Incidence of Celiac Disease in the United Kingdom: A Population-based Cohort Study. *Epidemiology (Cambridge, Mass)* 2019; 30(4): e23-e4.

56. West J, Fleming KM, Tata LJ, Card TR, Crooks CJ. Incidence and prevalence of celiac disease and dermatitis herpetiformis in the UK over two decades: population-based study. *Am J Gastroenterol* 2014; 109(5): 757-68.

57. Floreani A, Restrepo-Jimenez P, Secchi MF, et al. Etiopathogenesis of autoimmune hepatitis. *J Autoimmun* 2018; 95: 133-43.

58. Lash TL, Cole SR. Immortal person-time in studies of cancer outcomes. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology* 2009; 27(23): e55-6.

59. Romeri E, Baker A, Griffiths C. Mortality by deprivation and cause of death in England and Wales, 1999-2003. *Health Stat Q* 2006; (32): 19-34.

60. Kanzler S, Lohr H, Gerken G, Galle PR, Lohse AW. Long-term management and prognosis of autoimmune hepatitis (AIH): a single center experience. *Z Gastroenterol* 2001; 39(5): 339-41, 44-8.

61. National Health Service U. https://nhsbtdbe.blob.core.windows.net/umbraco-assetscorp/12250/nhsbt-liver-transplantation-annual-report-2017-2018.pdf, access date 15 May 2019.

#### **FIGURE LEGENDS**

Figure 1. Sex- and age-specific incidence rates of autoimmune hepatitis in England, 1997-2015.

**Figure 2. Sex and age- standardised incidence rates (SIR) of autoimmune hepatitis in England for each calendar year 1997-2015.** Incidence rates were standardised with the direct method to the European standard population's sex and age distribution 2013.

**Figure 3. Stacked curve of cause-specific cumulative incidence of death following autoimmune hepatitis diagnosis.** Estimates were based on the competing risk estimator of cumulative incidence of death with other causes of death as competing risks.